Author's response to reviews

Title: Improving educational achievement and anaemia of school children: design of a cluster randomised trial of school-based malaria prevention and enhanced literacy instruction in Kenya

Authors:

Simon Brooker (simon.brooker@lshtm.ac.uk)
George Okello (gokello@nairobi.kemri-wellcome.org)
Kiambo Njagi (knjagi@domckeny.or.ke)
Margaret M Dubeck (margaret_dubeck@harvard.edu)
Katherine E Halliday (katehalliday5@gmail.com)
Hellen Inyega (hinyega@yahoo.com)
Matthew CH Jukes (jukesma@gse.harvard.edu)

Version: 5 Date: 6 October 2010

Author's response to reviews:

References: reference 36, in preparation. Articles that have not been accepted for publication should not be cited in the reference list, unless they are publicly available as a pre-print. Instead, please refer to them in the text in the following way: (Authors: Title of manuscript, in preparation). Please ensure all other citations are re-numbered accordingly.

- We have removed ref 36, amended the reference details accordingly and included the reference in the text.

Figures: All figures must be cited in consecutive numerical order. Currently the figure citations are out of sequence. Please amend this.

- The order of figures has been corrected.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

- Checked